Author's response to reviews

Title: Does abdominal fat accumulation influence health and health related quality of life in primary school children? First cross-sectional results from the Baden-Wuerttemberg Study

Authors:

Dorothea Keszyues (dorothea.kesztyues@uni-ulm.de)
Tamara Wirt (tamara.wirt@uniklinik-ulm.de)
Susanne Kobel (susanne.kobel@uni-ulm.de)
Anja Schreiber (anja.schreiber@uni-ulm.de)
Sarah Kettner (sarah.kettner@uniklinik-ulm.de)
Clemens Drenowatz (clemens.drenowatz@uniklinik-ulm.de)
Jens Dreyhaupt (jens.dreyhaupt@uni-ulm.de)
Reinhold Kilian (reinhold.kilian@bkh-guenzburg.de)
Jürgen M Steinacker (juergen.steinacker@uni-ulm.de)

Version: 3 Date: 31 May 2012

Author's response to reviews:

1. Ethical approvement and trial registry number added in section "methods"
2. Competing interests statement added